Author’s response to reviews

Title: Inequities in incidence, morbidity and expenditures on prevention and treatment of malaria in southeast Nigeria Nkolika Uguru * 1, 5, Obinna Onwujeke 2, 5, Benjamin Uzochukwu 3, 5, Godwin Igiliegbe 4 and Soludo Eze 5

1Department of Preventive Dentistry, Faculty of Dentistry, College of Medicine, University of Nigeria, Enugu, Nigeria; Email; nk_uguru@yahoo.com 2Department of Health Administration and Management, Faculty of Health Sciences and Technology, College of Medicine, University of Nigeria, Enugu, Nigeria Email; onwujekwe@yahoo.co.uk 3Department of Community Medicine, University of Nigeria, Enugu, Nigeria Email; bscuzochukwu@yahoo.com 4Malaria and Vector Control Unit, Enugu State Ministry of Health, Enugu, Nigeria Email; igiliegbechuks@yahoo.com 5Health Policy Research Group, Department of Pharmacology and Therapeutics, College of Medicine, University of Nigeria, Enugu, Nigeria Email;

Authors:

Nkolika P Uguru (nk_uguru@yahoo.com)
Obinna E Onwujeke (onwujekwe@yahoo.co.uk)
Benjamin S Uzochukwu (bscuzochukwu@yahoo.com)
Godwin C Igiliegbe (igiliegbechuks@yahoo.com)
Soludo B Eze (soludo2001@yahoo.co.uk)

Version: 6 Date: 16 August 2009

Author’s response to reviews:

Dear Sir,

This is in reply to the mail sent to me on the 13th of August 2009 by the editorial board in which I was told to resend my title page and Abstract which were inadvertently missing from the last submission. This is not a revision of the manuscript and so the previous covering letter still stands.

Sincerely,

Dr NP Uguru